Author’s response to reviews

Title: Effect of whey protein supplementation on levels of endocannabinoids and some of metabolic risk factors in obese women on a weight-loss diet: A study protocol for a randomized controlled trial

Authors:

Fatemeh Haidari (haidari58@gmail.com)
Vahideh Aghamohammadi (v_agamohammadi@yahoo.com)
Majid Mohammadshahi (mohamadshahi-m@ajums.ac.ir)
Kambiz Ahmadi (kzfir@yahoo.com)

Version: 1 Date: 22 Jun 2017

Author’s response to reviews:

Editorial Office; Nutrition Journal

We regarded the change that you requested.

We correct the our document according to Submission Guidelines:

1. Added the title page and Declarations section including all of the subheadings and figure legends at the end of the document.

2. We divided the document into the following sections (in this order):

   • Title page
   • Abstract
   • Keywords
   • Main text
   • List of abbreviations
   • Declarations
     o Ethics approval and consent to participate
     o Consent for publication
We provided Figures as separate files.

The revised sections are yellow color and highlight.

We confirm the names and order of authors.